Author's response to reviews

Title: Exploring evidence-policy linkages in health research plans: a case study from six countries

Authors:

- Shamsuzzoha B Syed (ssyed@jhsph.edu)
- Adnan A Hyder (ahyder@jhsph.edu)
- Gerald Bloom (G.Bloom@ids.ac.uk)
- Sandhya Sundaram (ssundara@jhsph.edu)
- Abbas Bhuiya (abbas@icddrb.org)
- Zhang Zhenzhong (zhangzzc@126.com)
- Barun Kanjilal (barun@iihmnr.org)
- Oladimeji Oladepo (oladepod@yahoo.com)
- George Pariyo (gpariyo@iph.ac.ug)
- David H Peters (dpeters@jhsph.edu)
- Future Health Systems: Innovations for Equity (fhs@jhsph.edu)

Version: 5 Date: 29 February 2008

Author's response to reviews:

Thank you for your email dated 2/23/08. We have made the requested formatting changes and are resubmitting a new version of the paper through the submission system.

Please let me know whether you need anything else.